% EJC papers *must* begin with the following two lines.
\documentclass[12pt]{article}
\usepackage{e-jc}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb,extarrows}
\usepackage{xspace}
% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}

% all overfull boxes must be fixed;
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\def\red{\textcolor{red} }
\def\l{\ell}
\def\w{w}
\def\u{\ensuremath{\mathcal U}\xspace}

\newtheorem{theorem}{Theorem}%[section]
\newtheorem{cor}[theorem]{Corollary}
\newtheorem{prop}[theorem]{Proposition}
\newtheorem{conj}[theorem]{Conjecture}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{ex}[theorem]{Example}
\newtheorem{problem}[theorem]{Problem}


\newcommand{\doublead}{{\rm doublead\,}}
\newcommand{\run}{{\rm run\,}}
\newcommand{\cpk}{{\rm cpk\,}}
\newcommand{\bs}{{\mathfrak S}}
\newcommand{\lpk}{{\rm lpk\,}}
\newcommand{\single}{{\rm single\,}}
\newcommand{\pk}{{\rm pk\,}}
\newcommand{\des}{{\rm des\,}}
\newcommand{\Exc}{{\rm Exc\,}}
\newcommand{\exc}{{\rm exc\,}}
\newcommand{\aexc}{{\rm aexc\,}}
\newcommand{\we}{{\rm wexc\,}}
\newcommand{\ce}{{\rm cexc\,}}
\newcommand{\cyc}{{\rm cyc\,}}
\newcommand{\fix}{{\rm fix\,}}
\newcommand{\dc}{{\rm dc\,}}
\newcommand{\DS}{\mathcal{DS}}
\newcommand{\mc}{{\mathcal C}_A}
\newcommand{\cc}{{\mathcal C}}
\newcommand{\man}{\mathfrak{A}_n}
\newcommand{\mdn}{\mathcal{D}}
\newcommand{\D}{\mathfrak{D}}
\newcommand{\msb}{\mathfrak{S}_8}
\newcommand{\msm}{\mathfrak{S}_m}
\newcommand{\mcn}{\mathfrak{C}_{2n}}
\newcommand{\mcnn}{\mathfrak{C}_{2n+1}}
\newcommand{\msn}{\mathfrak{S}_n}
\newcommand{\rss}{\mathcal{SS}}
\newcommand{\ms}{\mathfrak{S}}
\newcommand{\inv}{{\rm inv\,}}
\newcommand{\msnn}{\mathfrak{S}_{n+1}}
\newcommand{\ud}{{\rm ud\,}}
\newcommand{\as}{{\rm as\,}}
\newcommand{\rs}{\mathcal{RS}}
\newcommand{\bn}{{\rm\bf n}}
\newcommand{\lrf}[1]{\lfloor #1\rfloor}
\newcommand{\lrc}[1]{\lceil #1\rceil}
\newcommand{\sep}{\preceq}
\newcommand{\sgn}{{\rm sgn\,}}
\newcommand{\mbn}{{\mathcal B}_n}
\newcommand{\mqn}{\mathcal{Q}_n}
\newcommand{\z}{ \mathbb{Z}}
\newcommand{\asc}{{\rm asc\,}}
\DeclareMathOperator{\N}{\mathbb{N}}
\DeclareMathOperator{\LHP}{LHP}
\DeclareMathOperator{\R}{\mathbb{R}}
\DeclareMathOperator{\Q}{\mathbb{Q}}
\DeclareMathOperator{\re}{Re}
\newcommand{\rz}{{\rm RZ}}
\newcommand{\Eulerian}[2]{\genfrac{<}{>}{0pt}{}{#1}{#2}}
\newcommand{\Stirling}[2]{\genfrac{\{}{\}}{0pt}{}{#1}{#2}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{\bf Context-free grammars for several polynomials associated with Eulerian polynomials}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address

\author{Shi-Mei Ma\thanks{Supported by NSFC (11401083), Natural Science Foundation of Hebei Province (A2017501007) and
the Fundamental Research Funds for the Central Universities (N152304006).}\\
\small School of Mathematics and Statistics\\[-0.8ex]
\small Northeastern University at Qinhuangdao\\[-0.8ex]
\small Hebei 066000, P.R. China\\
\small\tt shimeimapapers@163.com\\
\and
Jun Ma\thanks{Supported by NSFC 11571235.}\\
\small Department of mathematics\\[-0.8ex]
\small Shanghai jiao tong university\\[-0.8ex]
\small Shanghai, P.R. china\\
\small\tt majun904@sjtu.edu.cn\\
\and
Yeong-Nan Yeh\thanks{Supported partially by NSC under the Grant No. 104-2115-M-001-010.}\\
\small Institute of Mathematics\\[-0.8ex]
\small Academia Sinica\\[-0.8ex]
\small Taipei, Taiwan\\
\small\tt mayeh@math.sinica.edu.tw\\
\and
Bao-Xuan Zhu\thanks{Supported by NSFC 11571150.}\\
\small School of Mathematical Sciences\\[-0.8ex]
\small Jiangsu Normal University\\[-0.8ex]
\small Xuzhou 221116, P.R. China\\
\small\tt bxzhu@jsnu.edu.cn\\
}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Apr 19, 2017}{, 2018}{, 2018}\\
\small Mathematics Subject Classifications: 05A05, 05A15}

\begin{document}
\begin{thebibliography}{24}
%\bibitem{Andre84}
%D. Andr\'e, \'Etude sur les maxima, minima et s\'equences des permutations, \newblock {\em Ann. Sci. \'Ecole Norm. Sup.}, 3(1) (1884), 121--135.

\bibitem{Bona12}
M. B\'ona, Combinatorics of Permutations, second edition, CRC Press, Boca Raton, FL, 2012.

\bibitem{Bra06}
P. Br\"and\'en, On linear transformations preserving the P\'olya
frequency property, \newblock {\em Trans. Amer. Math. Soc.}, 358 (2006), 3697--3716.

\bibitem{Brenti90}
F. Brenti, Unimodal polynomials arising from symmetric functions, \newblock {\em Proc. Amer.
Math. Soc.}, 108 (1990), 1133--1141.


\bibitem{Bre94}
F. Brenti, $q$-Eulerian polynomials arising from Coxeter groups, \newblock {\em European J. Combin.}, 15 (1994), 417--441.

\bibitem{Bre00}
F. Brenti, A class of $q$-symmetric functions arising from plethysm,
\newblock {\em J. Combin. Theory Ser. A}, 91 (2000), 137--170.

\bibitem{Chen93}
W.Y.C. Chen, Context-free grammars, differential operators and formal
power series, \newblock {\em Theoret. Comput. Sci.}, 117 (1993), 113--129.

\bibitem{Chen09}
W.Y. C. Chen, R.L. Tang and A.F. Y. Zhao, Derangement polynomials and excedances of type $B$,
\newblock {\em Electron. J. Combin.}, 16(2) (2009), \#R15.


\bibitem{Chen16}
W.Y.C. Chen, A.M. Fu, Context-free grammars for permutations and increasing trees, \newblock {\em Adv. in Appl. Math.}, 82 (2017), 58--82.

\bibitem{Chow09}
C.-O. Chow, On derangement polynomials of type $B$. II, \newblock {\em J. Combin. Theory Ser.
A}, 116 (2009), 816--830.

\bibitem{Chow14}
C.-O. Chow, S-M. Ma, Counting signed permutations by their alternating runs, \newblock {\em Discrete Math.}, 323 (2014), 49--57.

%\bibitem{Conrad06}
%E.V.F. Conrad, P. Flajolet, The Fermat cubic, elliptic functions, continued fractions, and a combinatorial excursion,
%S\'em. Lothar. Combin, 54 (2006), Article B54g.

\bibitem{DB62}
F. N. David and D. E. Barton, Combinatorial Chance, Charles Griffin
and Company, Ltd. London, UK, 1962.

\bibitem{Dumont96}
D. Dumont, Grammaires de William Chen et d\'erivations dans les arbres et
arborescences, \newblock {\em S\'em. Lothar. Combin.}, 37, Art. B37a (1996), 1--21.


\bibitem{Dumont9602}
D. Dumont, A. Ramamonjisoa, Grammaire de Ramanujan et Arbres de Cayley, \newblock {\em Electron. J. Combin.}, 3 (1996), R17.

%\bibitem{Flajolet06}
%P. Flajolet, P. Dumas, and V. Puyhaubert, Some exactly solvable models
%of urn process theory, DMTCS proc. AG, 2006, 59--118.

%\bibitem{FS70}
%D. Foata, M. Sch\"utzenberger, Th\'eorie G\'eom\'etrique des
%Polyn\^omes Euleriens, Lecture Notes in Mathematics, vol. 138,
%Springer-Verlag, Berlin-New York, 1970.


%\bibitem{Kim01}
%D. Kim, J. Zeng, A new decomposition of derangements, \newblock {\em J. Combin. Theory Ser. A}, 96 (2001), 192--198.

\bibitem{Lin15}
Z. Lin, J. Zeng, The $\gamma$-positivity of basic Eulerian polynomials via group actions, \newblock {\em J. Combin. Theory Ser.
A}, 135 (2015), 112--129.

%\bibitem{Ma121}
%S.-M. Ma, Derivative polynomials and enumeration of permutations by number of interior and left peaks,
%\newblock {\em Discrete Math.}, 312 (2012), 405--412.

\bibitem{Ma131}
S.-M. Ma, Some combinatorial arrays generated by context-free grammars, \newblock {\em European J. Combin.},
34 (2013), 1081--1091.

\bibitem{Ma132}
S.-M. Ma, Enumeration of permutations by number of alternating runs, \newblock {\em Discrete Math.}, 313 (2013), 1816--1822.
%
\bibitem{Ma08}
S.-M. Ma, Y. Wang, $q$-Eulerian polynomials and polynomials with only real zeros, \newblock {\em Electron. J. Combin.}, 15 (2008), R17.

\bibitem{MaYeh}
S.-M. Ma, Y.-N. Yeh, Stirling permutations, cycle structure of
permutations and perfect matchings, \newblock {\em Electron. J. Combin.} 22(4) (2015), \#P4.42.


\bibitem{Ma16}
S.-M. Ma, Y.-N. Yeh, Eulerian polynomials, Stirling
permutations of the second kind and perfect matchings, \newblock {\em Electron. J. Combin.} 24(4) (2017), \#P4.27.


%\bibitem{Purtill93}
%M. Purtill, Andr\'e permutations, lexicographic shellability and the cd-index of a convex polytope, Trans. Amer. Math. Soc. 338 (1993) 77--104 .
%\bibitem{Pak15}
%I. Pak, R. Pemantle, On the longest $k$-alternating subsequence, \newblock {\em Electron. J. Combin.} 22(1)
%(2015), \#P1.48.

%\bibitem{MacMahon1915}
%P.A. MacMahon, Combinatory Analysis, 2 volumes, Cambridge University Press, London,
%1915--1916. Reprinted by Chelsea, New York, 1960.

%\bibitem{Zeng12}
%H. Shin, J. Zeng, The symmetric and unimodal expansion of Eulerian polynomials via continued fractions, \newblock {\em European J. Combin.}, 33 (2012), 111--127.
\bibitem{Shareshian07}
J. Shareshian, M.L. Wachs, $q$-Eulerian polynomials: excedance number and major index, \newblock {\em Electron. Res. Announc. Amer.
Math. Soc.}, 13 (2007), 33--45.

%\bibitem{Sloane}
%N.J.A. Sloane, The On-Line Encyclopedia of Integer Sequences,
%published electronically at
%http://oeis.org, 2010.

\bibitem{Sta08}
R.P. Stanley, Longest alternating subsequences of permutations,
\newblock {\em Michigan Math. J.}, 57 (2008), 675--687.

\bibitem{Stembridge92}
J.R. Stembridge, Eulerian numbers, tableaux, and the Betti numbers of a toric variety, \newblock {\em Discrete Math.},
99 (1992), 307--320.
%\bibitem{Springer}
%T.A. Springer, Remarks on a combinatorial problem, \newblock {\em Nieuw Arch. Wisk.} 19 (1971), 30--36.
%
%\bibitem{Sta08}
%R.P. Stanley, Longest alternating subsequences of permutations,
%\newblock {\em Michigan Math. J.} 57 (2008), 675--687.
%
%\bibitem{Sundaram1994}
%S. Sundaram, The homology representations of the symmetric group on Cohen-Macaulay subposets of the partition lattice, \newblock {\em Adv. Math.} 104 (1994), 225--296.
%
%\bibitem{Sundaram1995}
%S. Sundaram, The homology of partitions with an even number of blocks, \newblock {\em J. Algebraic Combin.} 4 (1995),
%69--92.
%
%\bibitem{Sundaram1996}
%S. Sundaram, Plethysm, partitions with an even number of blocks and Euler numbers, DIMACS Series
%in Discrete Mathematics and Theoretical Computer Science 24, (Billera, Greene, Simion,
%Stanley, eds.), Amer. Math. Soc. (1996), 171--198.
%\bibitem{Bona12}
%M. B\'ona,, Combinatorics of Permutations, Second Edition, Chapman \& Hall/CRC, Boca Raton, Florida, 2012.
%\bibitem{Janson11}
%S. Janson, M. Kuba and A. Panholzer, Generalized Stirling permutations, families of increasing trees and urn models,
%\newblock {\em J. Combin. Theory Ser. A,} 118 (2011) 94--114.
\bibitem{Zhang95}
X.D. Zhang, On $q$-derangement polynomials, \newblock {\em Combinatorics and Graph Theory} 95,
Vol.~1 (Hefei), World Sci. Publishing, River Edge, NJ, 1995, pp. 462--465.

\bibitem{Zhao11}
A.F.Y. Zhao, The combinatorics on permutations and derangements of type $B$, \newblock Ph.D. dissertation, Nankai University, 2011.
\end{thebibliography}


\maketitle

\begin{abstract}
In this paper, we present grammatical descriptions of several polynomials associated with Eulerian polynomials,
including $q$-Eulerian polynomials, alternating run polynomials and derangement polynomials. As applications, we get
several convolution formulas involving these polynomials.
  % keywords are optional

\bigskip\noindent
\textbf{Keywords:} Eulerian polynomials; Alternating runs; Derangement polynomials; Context-free grammars
\end{abstract}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Introduction}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%In the following, we recall some definitions, notation and results.
For an alphabet $A$, let $\mathbb{Q}[[A]]$ be the rational commutative ring of formal power
series in monomials formed from letters in $A$. Following~\cite{Chen93}, a context-free grammar over
$A$ is a function $G: A\rightarrow \mathbb{Q}[[A]]$ that replaces a letter in $A$ by a formal function over $A$.
The formal derivative $D$ is a linear operator defined with respect to a context-free grammar $G$. More precisely,
the derivative $D=D_G$: $\mathbb{Q}[[A]]\rightarrow \mathbb{Q}[[A]]$ is defined as follows:
for $x\in A$, we have $D(x)=G(x)$; for a monomial $u$ in $\mathbb{Q}[[A]]$, $D(u)$ is defined so that $D$ is a derivation,
and for a general element $q\in\mathbb{Q}[[A]]$, $D(q)$ is defined by linearity. %Therefore,
%for any formal functions $u$ and $v$,
%it follows from the {\it Leibniz's formula} that
%\begin{equation*}\label{Dnab-Leib}
%D^n(uv)=\sum_{k=0}^n\binom{n}{k}D^k(u)D^{n-k}(v).
%\end{equation*}

%In combinatorics, the study of context-free grammars began in 1993~\cite{Chen93} and there has been much recent activity.
Many combinatorial structures can be generated by using context-free
grammars, such as set partitions~\cite{Chen93}, permutations~\cite{Dumont96,Ma132},
Stirling permutations~\cite{Chen16,Ma16}, increasing trees~\cite{Chen16,Dumont96},
rooted trees~\cite{Dumont9602} and perfect matchings~\cite{Ma131}.
In this paper, using
a grammatical labeling introduced by Chen and Fu~\cite{Chen16}, we
study several polynomials associated with the Eulerian polynomials.

Let $\msn$ be the symmetric group on the set $[n]=\{1,2,\ldots,n\}$, and
let $\pi=\pi(1)\cdots\pi(n)\in\msn$. We say that
the index $i\in [n-1]$ is an {\it excedance} of $\pi$ if $\pi(i)>i$.
%let $i$ be called an {\it anti-excedance} of $\pi$ if $\pi(i)<i$, and let $i$ be called a {\it fixed point} of $\pi$ if $\pi(i)=i$.
Denote by $\exc(\pi)$ the number of excedances of $\pi$. The {\it Eulerian polynomials} are defined by
\begin{equation*}
A_0(x)=1,~A_n(x)=\sum_{\pi\in\msn}x^{\exc(\pi)}=\sum_{k=0}^{n-1}\Eulerian{n}{k}x^{k}~\textrm{for $n\ge 1$},
\end{equation*}
where $\Eulerian{n}{k}$ are the {\it Eulerian numbers}.
The numbers $\Eulerian{n}{k}$
satisfy the recurrence relation
$$\Eulerian{n}{k}=(k+1)\Eulerian{n-1}{k}+(n-k)\Eulerian{n-1}{k-1},$$
with the initial conditions $\Eulerian{0}{0}=1$ and $\Eulerian{0}{k}=0$ for $k\geq 1$.

The {\it hyperoctahedral group} $B_n$ is the group of signed permutations of the set $\pm[n]$ such that $\pi(-i)=-\pi(i)$ for all $i$, where $\pm[n]=\{\pm1,\pm2,\ldots,\pm n\}$. Throughout this paper, we always identify a signed permutation $\pi=\pi(1)\cdots\pi(n)$ with the word $\pi(0)\pi(1)\cdots\pi(n)$, where $\pi(0)=0$.
Let $$\des_B(\pi)=\#\{i\in\{0,1,2,\ldots,n-1\}|\pi(i)>\pi({i+1})\}.$$ The {\it Eulerian polynomials of type $B$} are defined by
$${B}_n(x)=\sum_{\pi\in B_n}x^{\des_B(\pi)}=\sum_{k=0}^nB(n,k)x^{k},$$
where $B(n,k)$ are called the {\it Eulerian numbers of type $B$}.
The numbers $B(n,k)$ satisfy
the recurrence relation
$$B(n+1,k)=(2k+1)B(n,k)+(2n-2k+3)B(n,k-1),$$
with the initial conditions $B(0,0)=1$ and $B(0,k)=0$ for $k\geq 1$.
Let us now recall two results on
context-free grammars.
\begin{prop}[{\cite[Section~2.1]{Dumont96}}]\label{Dumont96}
If $A=\{x,y\}$ and $G=\{x\rightarrow xy, y\rightarrow xy\}$,
then for $n\ge 1$
\begin{equation*}
D^n(x)=x\sum_{k=0}^{n-1}\Eulerian{n}{k}x^{k}y^{n-k}.
\end{equation*}
\end{prop}

\begin{prop}[{\cite[Theorem~10]{Ma131}}]\label{Ma13}
If $A=\{x,y\}$ and $G=\{x\rightarrow xy^2, y\rightarrow x^2y\}$,
then for $n\geq 1$,
$$D^n(x^2)=2^n\sum_{k=0}^{n-1}\Eulerian{n}{k}x^{2n-2k}y^{2k+2},$$
$$D^n(xy)=\sum_{k=0}^nB(n,k)x^{2n-2k+1}y^{2k+1}.$$
\end{prop}

The {\it $q$-Eulerian polynomials} of types $A$ and $B$ are respectively defined by
$$A_n(x;q)=\sum_{\pi\in\msn}x^{\exc(\pi)}q^{\cyc(\pi)},$$
$$B_n(x;q)=\sum_{\pi\in B_n}x^{\des_B(\pi)}q^{N(\pi)},$$
where $\cyc(\pi)$ is the number of cycles in $\pi$ and $N(\pi)=\#\{i\in [n]: \pi(i)<0\}$.
Following~\cite[157-162]{DB62} and~\cite[Theorem~4.3.3]{Zhao11},
the {\it alternating run polynomials} of types $A$ and $B$ can be respectively defined by
\begin{align*}
R_n(x)&=(1-w)\left(\dfrac{1+x}{2}\right)^{n-1}(1+w)^{n}A_n\left(\dfrac{1-w}{1+w}\right),\\
T_n(x)&=\frac{x}{2}\left(\dfrac{1+x}{2}\right)^{n-1}(1+w)^{n}B_n\left(\dfrac{1-w}{1+w}\right)
\end{align*}
for $n\geq 2$,
where $w=\sqrt{\frac{1-x}{1+x}}$.
Following~\cite[Proposition~5]{Brenti90} and~\cite[Theorem~3.2]{Chow09},
the {\it derangement polynomials} of types $A$ and $B$ can be respectively defined by
$$d_n(x)=\sum_{k=0}^n(-1)^{n-k}\binom{n}{k}A_k(x),$$
$$d_n^B(x)=\sum_{k=0}^n(-1)^{n-k}x^{n-k}\binom{n}{k}B_k(x).$$

The purpose of this paper is to explore grammatical descriptions of
the $q$-Eulerian polynomials, the alternating run polynomials and the derangement polynomials.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{$q$-Eulerian polynomials}\label{Section:02}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
According to~\cite[Proposition 7.2]{Bre00}, the polynomials $A_n(x;q)$ satisfy the recurrence relation
\begin{equation}\label{anxq-rr}
A_{n+1}(x;q)=(nx+q)A_{n}(x;q)+x(1-x)\frac{\partial}{\partial x}A_{n}(x;q),
\end{equation}
with the initial condition $A_0(x;q)=1$.
In~\cite[Theorem~3.4]{Bre94}, Brenti showed that the polynomials $B_n(x;q)$ satisfy the recurrence relation
\begin{equation}\label{bnxq-rr}
B_{n+1}(x;q)=((n+nq+q)x+1)B_{n}(x;q)+(1+q)x(1-x)\frac{\partial}{\partial x}B_{n}(x;q),
\end{equation}
with the initial condition $B_0(x;q)=1$.
The polynomials $A_n(x;q)$ and $B_n(x;q)$ have been extensively studied (see~\cite{Bra06,Ma08} for instance).
\subsection{Context-free grammar for $A_n(x;q)$}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\hspace*{\parindent}

A {\it standard cycle
decomposition} of $\pi\in\mathfrak{S}_n$ is defined by requiring that each cycle is
written with its smallest element first, and the cycles are
written in increasing order of their smallest element. In the following discussion, we always write $\pi$ in standard cycle decomposition.
We say that the index $i\in [n]$ is an {\it anti-excedance} of $\pi$ if $\pi(i)\leq i$.
Denote by $\aexc(\pi)$ the number of anti-excedances of $\pi$.
It is clear that $\exc(\pi)+\aexc(\pi)=n$.
We now give a labeling of $\pi$ as follows:
\begin{itemize}
  \item [\rm ($i$)]Put a superscript label $z$ right after each excedance;
  \item [\rm ($ii$)]Put a superscript label $y$ right after each anti-excedance;
  \item [\rm ($iii$)]Put a superscript label $q$ before each cycle of $\pi$;
  \item [\rm ($iv$)]Put a superscript label $x$ at the end of $\pi$.
\end{itemize}
For example,
the permutation $(1,3,4)(2)(5,6)$ can be labeled as
$^q(1^z3^z4^y)^q(2^y)^q(5^z6^y)^x$. The weight of $\pi$ is the product of its labels.

Let $\msn(i,j,k)=\{\pi\in\msn: \aexc(\pi)=i, \exc(\pi)=j, \cyc(\pi)=k\}$.
When $n=1$, we have $\ms_1(1,0,1)=\{^q(1^y)^x\}$. When $n=2$, we have
$\ms_2(2,0,2)=\{^q(1^y)^q(2^y)^x\}$ and $\ms_2(1,1,1)=\{^q(1^z2^y)^x\}$.
Suppose we get all labeled permutations in $\msn(i,j,k)$ for all $i,j,k$.
Let $\pi'\in\ms_{n+1}$ be obtained from $\pi\in\msn(i,j,k)$ by inserting the entry $n+1$ into $\pi$.
We distinguish the following three cases:
\begin{itemize}
  \item [\rm ($c_1$)] If the entry $n+1$ is inserted as a new cycle $(n+1)$, then $\pi'\in\ms_{n+1}(i+1,j,k+1)$.
  In this case, the insertion of $n+1$ corresponds to the operation $x\rightarrow qxy$;
\item [\rm ($c_2$)]If the entry $n+1$ is inserted right after an excedance, then $\pi'\in\ms_{n+1}(i+1,j,k)$.
In this case, the insertion of $n+1$ corresponds to the operation $z\rightarrow yz$;
\item [\rm ($c_3$)]If the entry $n+1$ is inserted right after an anti-excedance, then $\pi'\in\ms_{n+1}(i,j+1,k)$.
In this case, the insertion of $n+1$ corresponds to the operation $y\rightarrow yz$.
\end{itemize}
By induction, we get the following result.
\begin{theorem}\label{thm:01}
If $A=\{x,y,z\}$ and $G=\{x\rightarrow qxy, y\rightarrow yz, z\rightarrow yz\}$,
then
\begin{equation}\label{Dnx-Euleiran}
D^n(x)=x\sum_{\pi\in\msn}y^{\aexc(\pi)}z^{\exc(\pi)}q^{\cyc(\pi)}.
\end{equation}
\end{theorem}
Setting $y=1$ in~\eqref{Dnx-Euleiran}, we get $$D^n(x)|_{y=1}=xA_n(z;q).$$
Furthermore, setting $y=z=1$ in~\eqref{Dnx-Euleiran}, we get
$$D^n(x)|_{y=z=1}=x\sum_{\pi\in\msn}q^{\cyc(\pi)}=xq(q+1)(q+2)\cdots (q+n-1).$$

\begin{cor}
For $n\geq 1$, we have
$$A_{n+1}(x;q)=qA_n(x;q)+qx\sum_{k=0}^{n-1}\binom{n}{k}A_k(x;q)A_{n-k}(x).$$
\end{cor}
\begin{proof}
Using the {\it Leibniz's formula},
%\begin{equation*}\label{Dnab-Leib}
%D^n(uv)=\sum_{k=0}^n\binom{n}{k}D^k(u)D^{n-k}(v),
%\end{equation*}
we get
$$D^{n+1}(x)=D^n(qxy)=qD^n(xy)=q\sum_{k=0}^n\binom{n}{k}D^k(x)D^{n-k}(y).$$
Combining Proposition~\ref{Dumont96} and Theorem~\ref{thm:01}, we obtain
$$A_{n+1}(x;q)=q\sum_{k=0}^n\binom{n}{k}A_k(x;q)x^{n-k}A_{n-k}\left(\frac{1}{x}\right).$$
Recall that $A_n(x)$ are symmetric, i.e., $x^nA_n\left(\frac{1}{x}\right)=xA_n(x)$ for $n\ge 1$. Thus we get the desired result.
\end{proof}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Context-free grammar for $B_n(x;q)$}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\hspace*{\parindent}

For a permutation $\pi\in B_n$, we define an {\it ascent} (resp. a descent) to be a position $i\in \{0,1,2\ldots,n-1\}$ such that
$\pi(i)<\pi(i+1)$ (resp. $\pi(i)>\pi(i+1)$). Let $\asc_B(\pi)$ be the number of ascents of $\pi\in B_n$. As usual, denote by $\overline{i}$
the negative element $-i$. Let $A=\{x,y,z,u\}$. We shall
show that the following grammar
\begin{equation}\label{grammar02}
G=\{x\rightarrow qxyu,y\rightarrow xyz,z\rightarrow yzu,u\rightarrow qxzu\}
\end{equation}
can be used to generate permutations of $B_n$.
Now we give a labeling of $\pi\in B_n$ as follows:
\begin{itemize}
  \item [\rm ($L_1$)]If $i$ is an ascent and $\pi(i+1)>0$, then put a superscript label $z$ and a subscript $x$ right after $\pi(i)$;
 \item [\rm ($L_2$)]If $i$ is a descent and $\pi(i+1)>0$, then put a superscript label $y$ and a subscript $u$ right after $\pi(i)$;
\item [\rm ($L_3$)]If $i$ is an ascent and $\pi(i+1)<0$, then put a superscript label $z$ and a subscript $qx$ right after $\pi(i)$;
\item [\rm ($L_4$)]If $i$ is a descent and $\pi(i+1)<0$, then put a superscript label $y$ and a subscript $qu$ right after $\pi(i)$;
\item [\rm ($L_5$)] Put a superscript label $y$ and a subscript $x$ at the end of $\pi$.
\end{itemize}
Note that the weight of $\pi$ is given by $$w(\pi)=xy(xz)^{\asc_B(\pi)}(yu)^{\des_B(\pi)}q^{N(\pi)}.$$

Let $$B_n(i,j,k)=\{\pi\in B_n: \asc_B(\pi)=i, \des_B(\pi)=j, N(\pi)=k\}.$$
When $n=1$, we have $B_1(1,0,0)=\{0^z_x1^y_x\}$ and $B_1(0,1,1)=\{0^y_{qu}\overline{1}^y_x\}$.
Note that
$$D(xy)=x^2yz+qxy^2u.$$ Thus the sum of weights of the elements of $B_1$ is given by $D(xy)$.

Suppose we get all labeled permutations in $B_n(i,j,k)$ for all $i,j,k$, where $n\geq 1$. Let
$\pi'\in B_{n+1}$ be obtained from $\pi\in B_n(i,j,k)$ by inserting the entry $n+1$ or $\overline{n+1}$.
We distinguish the following six cases:
\begin{itemize}
  \item [\rm ($c_1$)] If $i$ is an ascent and we insert $n+1$ right after $\pi(i)$, then $\pi'\in B_{n+1}(i,j+1,k)$. In this case,
   the insertion of $n+1$ corresponds to applying the rule $z\rightarrow yzu$ to the label $z$ associated with $\pi(i)$;
\item [\rm ($c_2$)]If $i$ is an ascent and we insert $\overline{n+1}$ right after $\pi(i)$, then $\pi'\in B_{n+1}(i,j+1,k+1)$.
  In this case, the insertion of $\overline{n+1}$ corresponds to applying the rule $x\rightarrow qxyu$ to the label $x$ associated with $\pi(i)$;
    \item [\rm ($c_3$)] If $i$ is a descent and we insert $n+1$ right after $\pi(i)$, then $\pi'\in B_{n+1}(i+1,j,k)$.
  In this case, the insertion of $n+1$ corresponds to applying the rule $y\rightarrow xyz$ to the label $y$ associated with $\pi(i)$;
\item [\rm ($c_4$)]If $i$ is a descent and we insert $\overline{n+1}$ right after $\pi(i)$, then $\pi'\in B_{n+1}(i+1,j,k+1)$.
  In this case, the insertion of $\overline{n+1}$ corresponds to applying the rule $u\rightarrow qxzu$ to the label $u$ associated with $\pi(i)$;
\item [\rm ($c_5$)]If we insert $n+1$ at the end of $\pi$, then $\pi'\in B_{n+1}(i+1,j,k)$.
  In this case, the insertion of $n+1$ corresponds to applying the rule $y\rightarrow xyz$ to the label $y$ at the end of $\pi$;
\item [\rm ($c_6$)]If we insert $\overline{n+1}$ at the end of $\pi$, then $\pi'\in B_{n+1}(i,j+1,k+1)$.
  In this case, the insertion of $\overline{n+1}$ corresponds to applying the rule $x\rightarrow qxyu$ to the label $x$ at the end of $\pi$.
\end{itemize}
%For example, the permutation $0\overline{3}12$ can be labeled as
%$0^{y}_{qu}\overline{3}^z_x1^z_x2^y_x$ and $qx^3y^2z^2u=xy(xz)^2(yu)q$ is its weight.

%To illustrate the relation between the action of the formal derivative $D$ of the grammar~\eqref{grammar02} and the insertion of $n+1$ or $\overline{n+1}$ into a permutation $\pi\in B_n$, we give the following example.
%Let $\pi=02\overline{1}4\overline{6}5\overline{7}~\overline{8}\in B_8$.
%The labeling of $\pi$ is given by
%$$0^z_x2^y_{qu}\overline{1}^z_x4^y_{qu}\overline{6}^z_x5^y_{qu}\overline{7}~^y_{qu}\overline{8}^y_x.$$
%If we insert 9 after 0, the resulting permutation is given below,
%$$0^z_x9^y_u2^y_{qu}\overline{1}^z_x4^y_{qu}\overline{6}^z_x5^y_{qu}\overline{7}~^y_{qu}\overline{8}^y_x.$$
%Thus the insertion of 9 after 0 corresponds to applying the rule $z\rightarrow yzu$ to the label $z$ associated with 0.
%If we insert $\overline{9}$ after 0, the resulting permutation is given below,
%$$0^y_{qu}\overline{9}^z_x2^y_{qu}\overline{1}^z_x4^y_{qu}\overline{6}^z_x5^y_{qu}\overline{7}~^y_{qu}\overline{8}^y_x.$$
%The insertion of $\overline{9}$ after 0 corresponds to applying the rule $x\rightarrow qxyu$ to the label $x$ associated with 0.
%If we insert 9 after 2, the resulting permutation is given below,
%$$0^z_x2^z_x9^y_{qu}\overline{1}^z_x4^y_{qu}\overline{6}^z_x5^y_{qu}\overline{7}~^y_{qu}\overline{8}^y_x.$$
%The insertion of $9$ after 2 corresponds to applying the rule $y\rightarrow xyz$ to the label $x$ associated with 2.
%If we insert $\overline{9}$ after 2, the resulting permutation is given below,
%$$0^z_x2^y_{qu}\overline{9}^z_{qx}\overline{1}^z_x4^y_{qu}\overline{6}^z_x5^y_{qu}\overline{7}~^y_{qu}\overline{8}^y_x.$$
%The insertion of $\overline{9}$ after 2 corresponds to applying the rule $u\rightarrow qxzu$ to the label $u$ associated with 2.
In general, the insertion of $n+1$ (resp. $\overline{n+1}$) into $\pi$ corresponds
to the action of the formal derivative $D$ on a superscript label (resp. subscript label).
By induction, we get the following result.
\begin{theorem}\label{thm-Bnxq}
If $D$ is the formal derivative with respect to the grammar~\eqref{grammar02},
then
\begin{equation}\label{Dnxy}
D^n(xy)=xy\sum_{\pi\in B_n}(xz)^{\asc_B(\pi)}(yu)^{\des_B(\pi)}q^{N(\pi)}.
\end{equation}
\end{theorem}
Setting $x=y=z=1$ in~\eqref{Dnxy}, we get $$D^n(xy)|_{x=y=z=1}=B_n(u;q).$$

A {\it perfect matching} of $[2n]$ is a partition of $[2n]$ into $n$ blocks of size $2$. Denote by $N({n,k})$ the number of perfect matchings of $[2n]$ with the restriction that only $k$ matching pairs have even larger entries.
The numbers $N(n,k)$ satisfy the recurrence relation
\begin{equation*}
N(n+1,k)=2kN(n,k)+(2n-2k+3)N(n,k-1)
\end{equation*}
for $n,k\geq 1$, where $N(1,1)=1$ and $N(1,k)=0$ for $k\geq 2$ or $k\leq 0$ (see~\cite[Proposition~1]{MaYeh}).
Let $N_n(x)=\sum_{k=1}^nN({n,k})x^k$. Set $N_0(x)=1$ and
$$\widetilde{N}_n(x)=x^{n}N_n\left(\frac{1}{x}\right)~\textrm{for $n\ge 0$}.$$
From~\cite[Corollary~11]{Ma131}, we have
\begin{equation}\label{NnxBnx}
B_n(x)=\sum_{k=0}^n\binom{n}{k}N_k(x)\widetilde{N}_{n-k}(x).
\end{equation}

We end this section by giving a dual of~\eqref{NnxBnx}.
By the same way as in the proof of Theorem~\ref{thm-Bnxq},
it is routine to check that if
$A=\{x,y,z,u\}$ and $G=\{x\rightarrow xyu,y\rightarrow xyz,z\rightarrow yzu,u\rightarrow xzu\}$,
then
\begin{align*}
D^n(x)&=u\sum_{k=0}^nN(n,k)(yu)^{k}(xz)^{n-k},\\
D^n(u)&=u\sum_{k=0}^nN(n,k)(xz)^{k}(yu)^{n-k},\\
D^n(xy)&=xy\sum_{k=0}^nB(n,k)(yu)^k(xz)^{n-k}.
\end{align*}
Note that $$D^{n+1}(x)=D^n(xyu)=\sum_{k=0}^n\binom{n}{k}D^k(u)D^{n-k}(xy).$$
Therefore, we get the following result.
\begin{prop}
For $n\geq 0$,
we have
\begin{equation}\label{NnxBnx02}
N_{n+1}(x)=x\sum_{k=0}^n\binom{n}{k}\widetilde{N}_k(x)B_{n-k}(x).
\end{equation}
\end{prop}

It should be noted that the convolution formulas~\eqref{NnxBnx} and~\eqref{NnxBnx02} reveal that there exist some connections between
Brauer algebra and Coxeter groups.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Alternating run polynomials}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\hspace*{\parindent}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\subsection{Grammar for alternating runs polynomials}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\hspace*{\parindent}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

For $\pi\in\msn$, a {\it left peak index} is an index $i\in[n-1]$ such that $\pi(i-1)<\pi(i)>\pi(i+1)$, where we take $\pi(0)=0$, and the entry $\pi(i)$ is called a {\it left peak}.
Let $P(n,k)$ be the number of permutations of $\msn$ with $k$ left peaks.
It is well known that the numbers $P(n,k)$ satisfy the recurrence relation
$$P({n,k})=(2k+1)P({n-1,k})+(n-2k+1)P({n-1,k-1}),$$
with the initial conditions $P(1,0)=1$ and $P(1,k)=0$ for $k\geq 1$ (see~\cite{Ma132} for instance).
We say that $\pi\in\msn$ changes
direction at position $i$ if either $\pi({i-1})<\pi(i)>\pi(i+1)$, or
$\pi(i-1)>\pi(i)<\pi(i+1)$, where $i\in\{2,3,\ldots,n-1\}$. We say that $\pi$ has $k$ {\it alternating
runs} if there are $k-1$ indices $i$ such that $\pi$ changes
direction at these positions. The {\it up-down runs} of a permutation $\pi$ are the alternating runs of $\pi$ endowed with a 0
in the front.
Let $R(n,k)$ (resp.~$M(n,k)$) be the number of permutations of $\msn$ with $k$ alternating runs (resp. up-down runs).
The numbers $R(n,k)$ and $M(n,k)$ respectively satisfy the recurrence relations
\begin{equation}\label{rnk-recurrence01}
R(n,k)=kR(n-1,k)+2R(n-1,k-1)+(n-k)R(n-1,k-2),
\end{equation}
\begin{equation}\label{rnk-recurrence02}
M(n,k)=kM(n-1,k)+M(n-1,k-1)+(n-k+1)M(n-1,k-2)
\end{equation}
for $n,k\ge 1$, where $R(1,0)=M(0,0)=M(1,1)=1$ and $R(1,k)=M(n,0)=M(0,k)=0$ for $n,k\ge 1$
(see~\cite{Ma132,Sta08}).
Let $R_n(x)=\sum_{k=1}^{n-1}R(n,k)x^k$ and $M_n(x)=\sum_{k=1}^{n}M(n,k)x^k$.
There is a unified grammatical descriptions of $R_n(x)$ and $M_n(x)$.
\begin{prop}[{\cite[Theorem~6]{Ma132}}]\label{Ma132}
If $A=\{x,y,z\}$ and
%\begin{equation}\label{grammar-run}
$G=\{x\rightarrow xy, y\rightarrow yz,z\rightarrow y^2\}$,
%\end{equation}
then
\begin{align*}
D^{n}(x^2)&=x^2\sum_{k=0}^{n}R(n+1,k)y^kz^{n-k},\\
D^n(x)&=x\sum_{k=1}^nM(n,k)y^kz^{n-k}.
\end{align*}
In particular, $D^{n}(x^2)|_{z=1}=x^2R_{n+1}(y),~D^n(x)|_{z=1}=xM_n(y)$.
\end{prop}

A {\it run} of a signed permutation $\pi\in B_n$ is defined as a maximal interval of consecutive elements on which the elements of $\pi$ are monotonic in the order $\cdots<\overline{2}<\overline{1}<0<1<2<\cdots$.
The {\it up signed permutations} are signed permutations with $\pi(1)> 0$.
For example, the up signed permutation $03\overline{1}24\overline{5}\in B_5$ has four runs, i.e., $03,3\overline{1},\overline{1}24$ and $4\overline{5}$.
Let $T(n,k)$ denote the number of up signed permutations in $B_n$ with $k$ alternating runs.
Zhao~\cite[Theorem 4.2.1]{Zhao11} showed that the numbers $T(n,k)$ satisfy the
the following recurrence relation
\begin{equation}\label{tnk-recurrence03}
T(n,k)=(2k-1)T(n-1,k)+3T(n-1,k-1)+(2n-2k+2)T(n-1,k-2)
\end{equation}
for $n\geqslant 2$ and $1\leqslant k\leqslant n$, where $T(1,1)=1$ and $T(1,k)=0$ for $k>1$.
The alternating run polynomials of type $B$ are defined by $T_n(x)=\sum_{k=1}^nT(n,k)x^k$.
The first few of the polynomials $T_n(x)$ are given as follows:
\begin{align*}
T_1(x)=x,
  T_2(x)=x+3x^2,
  T_3(x)=x+12x^2+11x^3,
  T_4(x)=x+39x^2+95x^3+57x^4.
\end{align*}

The similarity of the recurrence relations~\eqref{rnk-recurrence01},~\eqref{rnk-recurrence02} and~\eqref{tnk-recurrence03} suggests the existence of a context-free grammar for $T_n(x)$.
We now present the third main result of this paper.
\begin{theorem}
If $A=\{x,y,z\}$ and $G=\{x\mapsto xy^2,y\mapsto yz^2, z\mapsto y^4z^{-1}\}$, then for $n\geq 1$,
\begin{equation*}
\begin{split}
D^n(x^3y)&=x^3y\sum_{k=1}^{n+1}T(n+1,k)y^{2k-2}z^{2n-2k+2},\\
D^n(xy)&=xy(y^2+z^2)\sum_{k=1}^{n}T(n,k)y^{2k-2}z^{2n-2k},\\
D^n(x^2)&=2^nx^2\sum_{k=0}^{n}M(n,k)y^{2k}z^{2n-2k},\\
D^n(x^2y^2)&=2^{n-1}x^2(y^2+z^2)\sum_{k=1}^{n}R(n+1,k)y^{2k}z^{2n-2k},\\
D^n(y^2)&=2^{n}y^2\sum_{k=0}^{\lrf{n/2}}P(n,k)y^{4k}z^{2n-4k}.
\end{split}
\end{equation*}
%$$D^n(x^3y)=x^3y\sum_{k=0}^{n+1}T(n+1,k)y^{2k-2}z^{2n-2k+2}.$$
%$$D^n(x^2)=2^nx^2\sum_{k=0}^{n}P(n,k)y^{2k}z^{2n-2k}.$$
\end{theorem}
\begin{proof}
We only prove the assertion for $D^n(x^3y)$ and the others can be proved in a similar way.
Note that $D(x^3y)=x^3y(z^2+3y^2)$ and $D^2(x^3y)=x^3y(z^4+12y^2z^2+11y^4)$.
we define $\widetilde{T}(n,k)$ by
\begin{equation}\label{Dnx-def}
D^n(x^3y)=x^3y\sum_{k=0}^{n+1}\widetilde{T}(n+1,k)y^{2k-2}z^{2n-2k+2}.
\end{equation}
Then $\widetilde{T}(2,0)=0,\widetilde{T}(2,1)=T(2,1)=1$ and $\widetilde{T}(2,2)=T(2,2)=3$.
It follows from~\eqref{Dnx-def} that
\begin{align*}
  D^{n+1}(x^3y)& =D(D^n(x^3y)) \\
              & =x^3y\sum_{k=1}^{n+1}(2k-1)\widetilde{T}(n+1,k)y^{2k-2}z^{2n-2k+4}+x^3y\sum_{k=1}^{n+1}3\widetilde{T}(n+1,k)y^{2k}z^{2n-2k+2}\\
              & +x^3y\sum_{k=1}^{n+1}(2n-2k+2)\widetilde{T}(n+1,k)y^{2k+2}z^{2n-2k}.
\end{align*}
Hence
\begin{equation}\label{Dnx3y}
\widetilde{T}(n+2,k)=(2k-1)\widetilde{T}(n+1,k)+3\widetilde{T}(n+1,k-1)+(2n-2k+6)\widetilde{T}(n+1,k-2).
\end{equation}
By comparing~\eqref{Dnx3y} with~\eqref{tnk-recurrence03}, we see that
the numbers $\widetilde{T}(n,k)$ satisfy the same recurrence relation and initial conditions as $T(n,k)$, so they agree.
\end{proof}

It follows from the {\it Leibniz's formula} that
\begin{align*}
  D^{n}(x^3y)& =\sum_{k=0}^n\binom{n}{k}D^{k}(xy)D^{n-k}(x^2)\\
              & =xyD^n(x^2)+\sum_{k=1}^n\binom{n}{k}D^{k}(xy)D^{n-k}(x^2).
\end{align*}
So the following corollary is immediate.
\begin{cor}
We have
$$T_{n+1}(x)=2^nxM_n(x)+(1+x)\sum_{k=1}^n2^{n-k}\binom{n}{k}T_k(x)M_{n-k}(x).$$
\end{cor}

By using the fact that
\begin{align*}
  D^{n}(x^2y^2)& =\sum_{k=0}^n\binom{n}{k}D^{k}(xy)D^{n-k}(xy)\\
              & =2xyD^n(xy)+\sum_{k=1}^{n-1}\binom{n}{k}D^{k}(xy)D^{n-k}(xy),
\end{align*}
we get a grammatical proof of the following result.
\begin{prop}[{\cite[Theorem~13]{Chow14}}]
For $n\geqslant 2$, we have
\begin{equation*}
2^{n-1}R_{n+1}(x)=2T_n(x)+\frac{1+x}{x}\sum_{k=1}^{n-1}\binom{n}{k}T_k(x)T_{n-k}(x).
\end{equation*}
\end{prop}

Let $P_n(x)=\sum_{k=0}^{\lrf{n/2}}P(n,k)x^k$. From $$D^{n}(x^2y^2)=\sum_{k=0}^n\binom{n}{k}D^{k}(x^2)D^{n-k}(y^2),$$
we obtain
\begin{equation*}\label{RnxTnxPnx}
(1+x)R_{n+1}(x)=2x\sum_{k=0}^n\binom{n}{k}M_k(x)P_{n-k}(x^2).
\end{equation*}
Recall that B\'ona~\cite[Section 1.3.2]{Bona12} obtained the following identity:
\begin{equation*}\label{TnxRnx}
M_n(x)=\frac{1}{2}(1+x)R_n(x)~\textrm{for $n\ge 2$}.
\end{equation*}
Therefore,
$$M_{n+1}(x)=x\sum_{k=0}^n\binom{n}{k}M_k(x)P_{n-k}(x^2).$$
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Derangement polynomials}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\hspace*{\parindent}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\subsection{Definitions, notation
%and results}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\hspace*{\parindent}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

There is a larger literature devoted to derangement polynomials (see~\cite{Brenti90,Chen09,Chow09,Lin15} for instance). We now recall some basic definitions, notation
and results.
We say that a permutation $\pi$ is a {\it derangement} if $\pi(i)\neq i$ for any $i\in [n]$.
Let $\mdn_n$ be the set of derangements of $\msn$.
The {\it derangement polynomials} are defined by
$$d_n(x)=\sum_{\pi\in\mdn_n}x^{\exc(\pi)}.$$
The polynomials $d_n(x)$ satisfy the following recurrence relation
$$d_{n+1}(x)=nx\left(d_{n}(x)+d_{n-1}(x)\right)+x(1-x)d_{n}'(x)$$
for $n\geq 1$, with the initial conditions $d_0(x)=1,d_1(x)=0$ (see~\cite{Zhang95}).
Brenti~\cite[Proposition 5]{Brenti90} showed that
\begin{equation*}\label{dxz}
\sum_{n\geq 0}d_n(x)\frac{z^n}{n!}=\frac{1-x}{e^{xz}-xe^z}.
\end{equation*}

For $\pi\in\msn$, we define
$$\fix(\pi)=\#\{i\in [n]: \pi(i)=i\},$$
$$\dc(\pi)=\#\{i\in [n]: \pi(i)<i\}.$$
Clearly, $\exc(\pi)+\fix(\pi)+\dc(\pi)=n$.
In~\cite[Section~2.2]{Dumont96}, Dumont proved that
if $A=\{x,y,z,e\}$ and $G=\{x\rightarrow xy, y\rightarrow xy,z\rightarrow xy, e\rightarrow ez\}$,
then
\begin{equation}\label{Dumont}
D^n(e)=e\sum_{\pi\in\msn}x^{\exc(\pi)}y^{\dc(\pi)}z^{\fix(\pi)}.
\end{equation}
In particular, setting $y=e=1,z=0$ in~\eqref{Dumont}, we get $D^n(e)|_{y=e=1,z=0}=d_n(x)$.

%In the rest of this section, we shall consider the type $B$ derangements polynomials $d_n^B(x)$.
In~\cite[p.~431]{Bre94}, Brenti introduced a definition of type $B$ weak excedance. Let $\pi\in B_n$.
We say that $i\in [n]$ is a {\it type $B$ weak excedance} of $\pi$ if $\pi(i)=i$ or $\pi(|\pi(i)|)>\pi(i)$.
Let $\we(\pi)$ be the number of weak excedances of $\pi$.
It follows from~\cite[Theorem~3.15]{Bre94} that $$B_n(x)=\sum_{\pi\in B_n}x^{\we(\pi)}.$$

A {\it fixed point} of $\pi\in B_n$ is an index $i\in [n]$ such that $\pi(i)=i$. A {\it derangement of type $B$} is a signed permutation $\pi\in B_n$ with no fixed points.
Let $\mdn_n^B$ be the set of derangements of $B_n$. Following~\cite{Chow09}, the {\it type $B$ derangement polynomials} $d_n^B(x)$ are defined by
$$d_0^B(x)=1,~d_n^B(x)=\sum_{\pi\in \mdn_n^B}x^{\we(\pi)}\quad\textrm{for $n\ge 1$}.$$
The first few of the polynomials $d_n^B(x)$ are given as follows:
\begin{align*}
d_1^B(x)=1,
d_2^B(x)=1+4x,
d_3^B(x)=1+20x+8x^2,
d_4^B(x)=1+72x+144x^2+16x^3.
\end{align*}
Chow~\cite[Theorem~3.2]{Chow09} proved that
$$\sum_{n\geq 0}d_n^B(x)\frac{z^n}{n!}=\frac{1-x}{e^{(2x-1)z}-xe^z}.$$
Moreover, the polynomials $d_n^B(x)$ satisfy the following recurrence relation
\begin{equation*}\label{dnbx-recu}
d_{n+1}^B(x)=2nx\left(d_n^B(x)+d_{n-1}^B(x)\right)+d_n^B(x)+2x(1-x)\frac{d}{dx}d_n^B(x).
\end{equation*}
for $n\geq 2$, with the initial conditions $d_0(x)=1,d_1(x)=0$ (see~\cite[Proposition~3.1]{Chow09}).
Chen et al.~\cite{Chen09} showed that the limiting distribution of the coefficients of $d_n^B(x)$ is normal.

Given $\pi\in \mdn_n^B$. Clearly, $$\we(\pi)=\#\{i\in [n]: \pi(|\pi(i)|)>\pi(i)\}.$$
We say that $i$ is an {\it anti-excedance} of $\pi$ if $\pi(|\pi(i)|)<\pi(i)$.
Let $\aexc(\pi)$ be the number of anti-excedances of $\pi$.
We say that $i$ is a singleton if $(\overline{i})$ is a cycle of $\pi$.
Let $\single(\pi)$ be the number of singletons of $\pi$.
Then
\begin{equation}\label{single-exc}
\we(\pi)+\aexc(\pi)+\single(\pi)=n.
\end{equation}

In the following discussion, we always write $\pi$
by using its \emph{standard cycle decomposition}, in which each
cycle is written with its largest entry last and the cycles are written in ascending order of their
last entry. For example, $\overline{3}51\overline{7}2\overline{6}~\overline{4}\in \mdn_7^B$ can be written as
$(\overline{6})(\overline{7},\overline{4})(\overline{3},1)(2,5)$.
Let $(c_1,c_2,\ldots,c_i)$ be a cycle in
standard cycle decomposition of $\pi$. We say that $c_j$ is an {\it
ascent} in the cycle if $c_j<c_{j+1}$, where $1\leq j<i$.
We say that $c_j$ is a {\it descent} in the cycle if $c_j>c_{j+1}$, where $1\leq j\leq i$ and we set $c_{i+1}=c_1$.
As pointed out by Chow~\cite[p.~819]{Chow09}, if $\pi\in\mdn_n^B$ with no singletons,
then $\we(\pi)$ equals the sum of the number of ascents in each cycle and $\aexc(\pi)$
equals the sum of the number of descents in each cycle.
Let $d(n,i,j)$ be the number of derangements of type $B$ with $i$ weak excedances and $j$ anti-excedances.

We can now conclude the fourth main result of this paper from the discussion above.
\begin{theorem}\label{thm-derangement}
Let $A=\{x,y,z,e\}$ and
\begin{equation}\label{grammar-derangement}
G=\{x\rightarrow xy^2,y\rightarrow x^2y, z\rightarrow x^2y^2z^{-3}, e\rightarrow ez^4\}.
\end{equation}
Then
\begin{equation*}\label{Dne-derangement}
D^n(e)=e\sum_{\pi\in \mdn_n^B}x^{2\we(\pi)}y^{2\aexc(\pi)}z^{4\single(\pi)}.
\end{equation*}
Equivalently,
\begin{equation}\label{Dne-derangement}
D^n(e)=e\sum_{i,j\geq0}d(n,i,j)x^{2i}y^{2j}z^{4(n-i-j)}.
\end{equation}
Setting $y=z=1$ in~\eqref{Dne-derangement}, we get $$D^n(e)|_{y=z=1}=ed_n^B(x^2).$$
\end{theorem}
\begin{proof}
Now we give a labeling of $\pi\in \mdn_n^B$ as follows:
\begin{itemize}
  \item [\rm ($L_1$)]If $i$ is a singleton, then put a superscript label $z^4$ right after $\overline{i}$;
 \item [\rm ($L_2$)]If $c_i$ is an ascent in a cycle, then put a superscript label $x^2$ right after $c_i$;
\item [\rm ($L_3$)]If $c_i$ is a descent in a cycle, then put a superscript label $y^2$ right after $c_i$;
\item [\rm ($L_4$)]Put a superscript label $e$ in the front of $\pi$.
\end{itemize}
The weight of $\pi$ is defined by $$w(\pi)=ex^{2\we(\pi)}y^{2\aexc(\pi)}z^{4\single(\pi)}.$$
Note that $\mdn_1^B=\{^e(\overline{1}^{z^4})\}$ and
$\mdn_2^B=\{^e(\overline{2}^{z^4})(\overline{1}^{z^4}),^e(\overline{1}^{x^2}2^{y^2}),^e(1^{x^2}2^{y^2}),
^e(\overline{2}^{x^2}\overline{1}^{y^2}),^e(\overline{2}^{x^2}1^{y^2})\}$.
Thus the weight of $^e(\overline{1}^{z^4})$ is given by $D(e)$ and the sum of weights of the permutations in $\mdn_2^B$ is
given by $D^2(e)$, since
$D(e)=ez^4$ and $D^2(e)=e(z^8+4x^2y^2)$.

To illustrate the relation between the action of the formal derivative $D$ of
the grammar~\eqref{grammar-derangement} and the insertion of $n+1$ or $\overline{n+1}$
into a permutation $\pi\in \mdn_n^B$, we give the following example.
Let $\pi=(\overline{6})(\overline{7},\overline{4})(\overline{3},1)(2,5)\in\mdn_7^B$.
Then $\pi$ can be labeled as
$$^e(\overline{6}^{z^4})(\overline{7}^{x^2}\overline{4}^{y^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}).$$
We distinguish the following four cases:
\begin{itemize}
  \item [\rm ($c_1$)]If we insert $\overline{8}$ as a new cycle, then the resulting permutation is given below,
$$^e(\overline{8}^{z^4})(\overline{6}^{z^4})(\overline{7}^{x^2}\overline{4}^{y^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}).$$
This case corresponds to applying the rule $e\rightarrow ez^4$ to the label $e$.
 \item [\rm ($c_2$)]If we insert $\overline{8}$ or $8$ into the cycle $(\overline{6})$, then the resulting permutations are given below,
$$^e(\overline{8}^{x^2}~\overline{6}^{y^2})(\overline{7}^{x^2}\overline{4}^{y^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}),$$
$$^e(\overline{6}^{x^2}~8^{y^2})(\overline{7}^{x^2}\overline{4}^{y^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}),$$
$$^e({6}^{x^2}~{8}^{y^2})(\overline{7}^{x^2}\overline{4}^{y^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}),$$
$$^e(\overline{8}^{x^2}~{6}^{y^2})(\overline{7}^{x^2}\overline{4}^{y^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}).$$
It should be noted that in the latter two permutations, we need to replace $\overline{6}$ by $6$.
This case corresponds to applying the rule $z\rightarrow x^2y^2z^{-3}$ to the label $z^4$, since $D(z^4)=4x^2y^2$.
\item [\rm ($c_3$)]If we insert $\overline{8}$ or $8$ right after $\overline{7}$, then the resulting permutations are given below,
$$^e(\overline{6}^{z^4})(\overline{7}^{y^2}~\overline{8}^{x^2}~\overline{4}^{y^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}),$$
$$^e(\overline{6}^{z^4})(\overline{7}^{x^2}~{8}^{y^2}~\overline{4}^{y^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}).$$
This case corresponds to applying the rule $x\rightarrow xy^2$ to the label $x^2$, since $D(x^2)=2x^2y^2$.
\item [\rm ($c_4$)]If we insert $\overline{8}$ or $8$ right after $\overline{4}$, then the resulting permutations are given below,
$$^e(\overline{6}^{z^4})(\overline{7}^{x^2}~\overline{4}^{y^2}~\overline{8}^{x^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}),$$
$$^e(\overline{6}^{z^4})(\overline{7}^{x^2}~\overline{4}^{x^2}8^{y^2})(\overline{3}^{x^2}1^{y^2})(2^{x^2}5^{y^2}).$$
This case corresponds to applying the rule $y\rightarrow x^2y$ to the label $y^2$, since $D(y^2)=2x^2y^2$.
\end{itemize}

In general,
the insertion of $n+1$ or $\overline{n+1}$ into $\pi\in\mdn_n^B$ corresponds
to the action of the formal derivative $D$ on a superscript label.
By induction, we see
that grammar~\eqref{grammar-derangement} generates all of the derangements of type $B$.
\end{proof}

Let $D$ be the formal derivative with respect to the grammar~\eqref{grammar-derangement}.
Clearly, $D(z^4)=4x^2y^2$.
It is easy to verify that
$$D^n(x^2y^2)=2^n\sum_{k=0}^n\Eulerian{n+1}{k}x^{2k+2}y^{2n-2k+2}.$$
Hence
\begin{equation}\label{Dnz4}
D^n(z^4)=4D^{n-1}(x^2y^2)=2^{n+1}\sum_{k=0}^{n-1}\Eulerian{n}{k}x^{2k+2}y^{2n-2k}~\textrm{for $n\ge 1$}.
\end{equation}
Note that
\begin{align*}
D^{n+1}(e)&=D^n(ez^4)\\
&=\sum_{k=0}^n\binom{n}{k}D^k(e)D^{n-k}(z^4)\\
&=z^4D^n(e)+\sum_{k=0}^{n-1}\binom{n}{k}D^k(e)D^{n-k}(z^4).
\end{align*}

Combining~\eqref{Dnz4} and Theorem~\ref{thm-derangement}, we get the following corollary.
\begin{cor}\label{dnxAnx}
For $n\geq 1$,
we have
$$d_{n+1}^B(x)=d_n^B(x)+x\sum_{k=0}^{n-1}2^{n-k+1}\binom{n}{k}d_k^B(x)A_{n-k}(x).$$
Set $d_n^B=d_n^B(1)$. We have $$d_{n+1}^B=d_n^B+n!\sum_{k=0}^{n-1}2^{n-k+1}\frac{d_k^B}{k!}.$$
\end{cor}

Moreover, from~\eqref{Dne-derangement}, the following result is immediate.
\begin{cor}
For $n\geq 0$,
we have
\begin{equation}\label{dnij-recu}
d(n+1,i,j)=d(n,i,j)+2id(n,i,j-1)+2jd(n,i-1,j)+4(n-i-j+2)d(n,i-1,j-1),
\end{equation}
with initial conditions $d(1,0,0)=1$ and $d(1,i,j)=0$ for $i\neq 0$ or $j\neq 0$.
\end{cor}

Let $$G(x,y;t)=\sum_{n\geq 0}\sum_{\pi\in\mdn_n^B}x^{\we(\pi)}y^{\aexc(\pi)}\frac{t^n}{n!}.$$
\begin{prop}\label{thm-EGF-derangement}
We have
\begin{equation}
G(x,y;t)=\frac{e^{(1-2x)t}}{1-\frac{x}{y-x}(e^{2(y-x)t}-1)}.
\end{equation}
\end{prop}
\begin{proof}
Let $d_n(x,y)=\sum_{i,j\geq 0}d(n,i,j)x^iy^j$. It follows from~\eqref{dnij-recu} that the polynomials $d_n(x,y)$ satisfy the recurrence relation
\begin{equation}\label{dnxy}
d_{n+1}(x,y)=(1+4nxy)d_n(x,y)+(2xy-4x^2y)\frac{\partial}{\partial x}d_n(x,y)+(2xy-4xy^2)\frac{\partial}{\partial y}d_n(x,y),
\end{equation}
with the initial conditions $d_0(x,y)=d_1(x,y)=1$.
By rewriting~\eqref{dnxy} in terms of the generating function $G$, we get
\begin{equation}\label{Gex-diff}
(1-4xyt)G_t=G+(2xy-4x^2y)G_x+(2xy-4xy^2)G_y.
\end{equation}
It is routine to check that $$\widetilde{G}(x,y;t)=\frac{e^{(1-2x)t}}{1-\frac{x}{y-x}(e^{2(y-x)t}-1)}$$ satisfies~\eqref{Gex-diff}.
Also, $\widetilde{G}(x,y;0)=1$ and $\widetilde{G}(0,y;t)=\widetilde{G}(x,0;t)=e^t$. Hence $G=\widetilde{G}$.
\end{proof}

Let $P(\pi)=\#\{i\in [n]: \pi(i)>0\}$. Then $N(\pi)+P(\pi)=n$.
It is not hard to check that
\begin{equation*}
\sum_{n\geq 0}\sum_{\pi\in\mdn_n^B}x^{\we(\pi)}y^{\aexc(\pi)}z^{\single(\pi)}u^{P(\pi)}v^{N(\pi)}\frac{t^n}{n!}
=\frac{e^{\left(vz-(u+v)x\right)t}}{1-\frac{x}{y-x}\left(e^{(u+v)(y-x)t}-1\right)}.
\end{equation*}

In the proof of Theorem~\ref{thm-derangement}, if we further put a superscript
label $q$ before each cycle of $\pi\in\mdn_n^B$, then we get the following result.
\begin{prop}
Let $A=\{x,y,z,e\}$ and
$$G=\{x\rightarrow xy^2,y\rightarrow x^2y, z\rightarrow x^2y^2z^{-3}, e\rightarrow qez^4\}.$$
Then
\begin{equation*}
D^n(e)=e\sum_{\pi\in \mdn_n^B}x^{2\we(\pi)}y^{2\aexc(\pi)}z^{4\single(\pi)}q^{\cyc(\pi)}.
\end{equation*}
\end{prop}

%%%%%%%%%%%%%%%%%%%%
\section{Concluding remarks}\label{Section-5}
%\hspace*{\parindent}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
In this paper, we present grammatical descriptions of several polynomials associated with the Eulerian polynomials.
There are various $q$-analogue of the Eulerian polynomials.
A challenging work is to study grammatical descriptions of several enumerative polynomials of
Euler-Mahonian statistics (see~\cite{Shareshian07} for instance).
The Robinson-Schensted-Knuth correspondence
associates with any permutation a pair of paths in a Young graph.
It would be interesting to encode Young tableaux by introducing a grammatical labeling.
One of the possible ways is based on the work of Stembridge~\cite[Section~4]{Stembridge92},
which established a connection between permutations and pairs of a marked tableau and a standard tableau
of the same shape.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection*{Acknowledgements}
This work was finished while Y.-N. Yeh was visiting the School of Mathematical Sciences, Dalian University of Technology, Dalian, P.R. China.
The authors thank the referee for his valuable suggestions which lead to a substantial improvement of the paper.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\end{document}
